Author’s response to reviews

Title: Association of stress, depression, and suicidal ideation with subjective oral health status and oral functions in Korean adults aged 35 years or more

Authors:

Young Sun Kim (yskim0821@cup.ac.kr)
Han-Na Kim (hnkim@cju.ac.kr)
Jung-Ha Lee (Jhlee86@pusan.ac.kr)
Se-Yeon Kim (secan00@naver.com)
Eun-Joo Jun (jsilver@pusan.ac.kr)
Jin-Bom Kim (jbomkim@hanmail.net)

Version: 2 Date: 04 May 2017

Author’s response to reviews:

Author Response to Editorial Comments

In accordance with BioMed Central editorial policies and formatting guidelines, all submissions to BMC Oral Health must have a Declarations section which includes the mandatory subsections listed below. Please refer to the journal’s Submission Guidelines web page for information regarding the criteria for each sub-section (https://bmcoralhealth.biomedcentral.com/)

Thank you so much for valuable comments. We rewrote and added the requirements of following declarations.


The revised manuscript and tables were checked and totally corrected by mother tongue.

Best regards,

Jin-Bom Kim